PEER-REVIEW REPORT

Name of journal: *World Journal of Clinical Cases*

Manuscript NO: 75433

Title: Perfectionism: Limitations and Directions for Future Research

Provenance and peer review: Invited Manuscript; Externally peer reviewed

Peer-review model: Single blind
PEER-REVIEW REPORT

Reviewer’s code: 05845620
Position: Peer Reviewer
Academic degree: PhD, RN
Professional title: Lecturer

SPECIFIC COMMENTS TO AUTHORS

1. The Title: The title and the content have different characteristics with each other.
   A title should be created that includes adolescents and depression.
   Authors’ response: thank you. We changed the title: “Perfectionism and mental health problems: Limitations and Directions for Future Research”

2. Abstract No mention is made of the research being evaluated in the abstract. (by Livazović G, Kuzmanović)
   Authors’ response: According journals guideline: we limited to cite any work in abstract

3. Key Words: Although depression is mentioned in the keywords, the article deals with social relations and technology addiction. Therefore, keywords are not suitable.
   Authors’ response: thank you. We agree with comment and added new keyword “Mental health problems”
PEER-REVIEW REPORT

Reviewer’s code: 04025443
Position: Editorial Board
Academic degree: MD, PhD
Professional title: Doctor, Senior Researcher
Reviewer’s Country/Territory: Russia

SPECIFIC COMMENTS TO AUTHORS

1. Dear authors! I read with interest your manuscript "Perfectionism: Limitations and Directions for Future Research" which is the Letter to the Editor. Manuscripts of this type usually imply a critical analysis of a previously published paper to originate further discussion. However, there is either no criticism, or it is not directly shown.

Authors’ response: we currently changed the abstract

2. Some language polishing is required.

Authors’ response: we send the manuscript to edit by professional website.
EDITORIAL OFFICE’S COMMENTS

Authors must revise the manuscript according to the Editorial Office’s comments and suggestions, which are listed below:

(1) Science editor:

1. This letter aims to explain limitations and directions for future research in perfectionism among adolescents, which is associated with emotional disorders and suicide. However, the title is not enough to summarize the main idea of the letter.

Authors’ response: thank you. We changed the title: “Perfectionism and mental health problems: Limitations and Directions for Future Research”

2. Further, no mention is made of the research being evaluated in the abstract. Thirdly, the keyword "depression" is not suitable because the article only deals with social relations and technology addiction.

Authors’ response: Thank you. We agree with comment and added new keyword “Mental health problems”

3. At last, manuscripts of this type usually imply a critical analysis of a previously published paper to originate further discussion.

Authors’ response: the current manuscript is letter to editor type. Such analysis must be considered in mini review or systematic review
(2) Company editor-in-chief:

I have reviewed the Peer-Review Report, the full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. However, the quality of the English language of the manuscript does not meet the requirements of the journal. Before final acceptance, the author(s) must provide the English Language Certificate issued by a professional English language editing company. Please visit the following website for the professional English language editing companies we recommend: [https://www.wjgnet.com/bpg/gerinfo/240](https://www.wjgnet.com/bpg/gerinfo/240).

Authors’ response: Thank you for kindly consideration. We sent the manuscript to AJE digital editing service.

Then, we evaluate the manuscript by the EDITAGE.

[https://pubsure.researcher.life/certificate/29f2fff3](https://pubsure.researcher.life/certificate/29f2fff3)